Author's response to reviews

Title: A gender perspective on factors that influence outdoor recreational physical activity among the elderly

Authors:

Katarina Sjögren (katarina.sjogren@med.lu.se)
Louise Stjernberg (louise.stjernberg@bth.se)

Version: 8 Date: 4 June 2010

Author's response to reviews: see over
Dear Editorial Production Team

Thank you for your editing suggestions and instructions. We have corrected the tables from your suggestions. Please do not hesitate to contact me if you have any questions regarding the manuscript.

Best regards,

Katarina Sjögren

Major revisions

Tables: Tables must be created using a Tables tool, and should not be linked files, figures, or generated with tabbed text. Tables must be divided into rows and columns and be oriented in a portrait page layout. Please ensure that text alignment in columns is clear and consistent. We recommend the MS Word auto-format template: simple 1. Please do not include vertical lines in your tables.

Response: We have corrected the tables and used MS Word auto-format.

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Tables- please remove the visible vertical lines from your tables

Response: We have removed the visible vertical lines from all the tables.

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

Response: We have checked the manuscript.